Author's response to reviews

Title: Quality measures for the care of patients with lateral epicondylalgia

Authors:

Francisco Minaya-Muñoz (franminaya@mvclinic.es)
Francesc Medina-i-Mirapeix (mirapeix@um.es)
Fermín Valera-Garrido (ferminvalera@mvclinic.es)

Version: 2 Date: 25 April 2013

Author's response to reviews: see over
Title: Quality measures for the care of patients with lateral epicondylalgia

Authors:

1.- Francisco Minaya-Muñoz
PT. MVL Clinic Physiotherapy Centre (Madrid, Spain). Fremap Hospital (Madrid, Spain). Faculty of Medicine, San Pablo CEU University (Madrid, Spain).

101 Hermanos García Noblejas St., 3A; 28037, Madrid, Spain.
franminaya@mvclinic.es

2.- Francesc Medina-i-Mirapeix
PT, PhD. Department of Physiotherapy, University of Murcia (Murcia, Spain).

Campus Mare Nostrum, Universidad de Murcia, 30100, Campus de Espinardo, Murcia, Spain.
mirapeix@um.es

3.- Fermín Valera-Garrido
PT, PhD. MVL Clinic Physiotherapy Centre (Madrid, Spain). Fremap Hospital (Madrid, Spain). Faculty of Medicine, San Pablo CEU University (Madrid, Spain).

39 Gardenias St., 2A; 28925, Alcorcon, Madrid, Spain.
ferminvalera@mvclinic.es

Name and address for correspondence

Francisco Minaya-Muñoz
101 Hermanos García Noblejas St., 3A
28037, Madrid, Spain
E-mail: franminaya@mvclinic.es
Fax number: +34 916464980
Telephone: +34 664086454

Version: 2

Date: 25 April 2013.

Reviewer’s report:
1.) Ethics statement: Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

The study design and methods were reviewed and approved by the Academic Review Committee of Quality Management Program at the Faculty of Medicine of University of Murcia (Murcia, Spain).